Author's response to reviews

Title: Polymorphisms in metabolic genes, their combination and interaction with tobacco smoke and alcohol consumption and risk of gastric cancer: a case-control study in an Italian population

Authors:

Stefania Boccia (sboccia@rm.unicatt.it)
Fakhredin A. Sayed-Tabatabaei (fakhredin@gmail.com)
Roberto Persiani (rpersiani@rm.unicatt.it)
Francesco Gianfagna (francescogianfagna@yahoo.it)
Stefano Rausei (s.rausei@libero.it)
Dario Arzani (dario.arzani@rm.unicatt.it)
Antonio La Greca (antonio.lagreca@rm.unicatt.it)
Domenico D'Ugo (ddugo@rm.unicatt.it)
Giuseppe La Torre (giuseppe.latorre@rm.unicatt.it)
Cornelia M. van Duijn (c.vanduijn@erasmusmc.nl)
Gualtiero Ricciardi (wricciardi@rm.unicatt.it)

Version: 4 Date: 19 October 2007

Author's response to reviews: see over
Dear Dr Edmunds,

Please find enclosed the original article MS 1321081475152364 after the second revision. Reviewer no. 1 accepted the first revised version without comments, while the second one (Dr. Sichel) actually pointed out at some minor corrections on CYP nomenclature.

In the present submitted version of the MS the CYP nomenclature has been completely revised and I hope now it is correct. Thanks to Dr. Sichel for his remarkable help.

All authors have read and approved the revised manuscript.

With many thanks for your assistance,

Stefania Boccia